Author's response to reviews

Title: Statistical techniques to construct assays for identifying likely responders to a treatment under evaluation from cell line genomic data

Authors:

Erich P Huang (erich.huang@nih.gov)
Jane Fridlyand (fridlyand.jane@gene.com)
Nicholas Lewin-Koh (lewin-koh.nicholas@gene.com)
Peng Yue (yue.peng@gene.com)
Xiaoyan Shi (shi.xiaoyan@gene.com)
David Dornan (dornan.david@gene.com)
Bart Burington (burington.bart@gene.com)

Version: 11 Date: 25 October 2010

Author's response to reviews:

October 25, 2010

Dear Editor,

Attached is a copy of the manuscript “Statistical techniques to construct assays for identifying likely responders to a treatment under evaluation from cell line genomic data” (Manuscript Number 4241051003512508). We have corrected some typographical errors we have found during our last read-through of the manuscript.

Thank you,

Erich Huang
Jane Fridlyand
Nicholas Lewin-Koh
Peng Yue
Xiaoyan Shi
David Dornan
Bart Burington